Author's response to reviews

Title: No significant effect on bone mineral density by high doses of vitamin D3 given to overweight subjects for one year

Authors:

Rolf Jorde (rolf.jorde@unn.no)  
Monica Sneve (monica.sneve@unn.no)  
Peter A Torjesen (PeterAbusdal.Torjesen@akersykehus.no)  
Yngve Figenschau (yngve.figenschau@unn.no)  
John-Bjarne Hansen (john.bjarne.hansen@unn.no)  
Guri Grimnes (guri.grimnes@unn.no)

Version: 4 Date: 5 January 2010

Author's response to reviews:

The Editor  
Nutrition Journal  
Tromsø January 5, 2010

Concerning MS: 7353579533057361, “No significant effect on bone mineral density by high doses of vitamin D3 given to overweight subjects for one year”, Jorde R et al.

Thank you for your e-mail from January 5, 2010.

We have made the following changes:

Major revision:
- the revision number is now mentioned in the abstract
- all authors have read and approved the manuscript which is now included in the Authors’ contributions.

Minor revisions:
- full details of reference 18 is now included

Concerning the figures, if further cropping is necessary I please ask you to do so.

I have once again checked the manuscript for typographical errors.

Best regards

Rolf Jorde